Author's response to reviews

Title: Listening to patients: using verbal data in the validation of the Aberdeen Measures of Impairment, Activity Limitation and Participation Restriction (Ab-IAP)

Authors:

  Jeremy Horwood (j.horwood@bristol.ac.uk)
  Beth Pollard (beth.pollard@abdn.ac.uk)
  Salma Ayis (salma.ayis@kcl.ac.uk)
  Teresa McIlvenna (Teresa.McIlvenna@ubht.nhs.uk)
  Marie Johnston (m.johnston@abdn.ac.uk)

Version: 5 Date: 14 July 2010

Author's response to reviews: see over
Dear Editorial Board

RE: MS: 5693482003579478

Thank you very much for your letter indicating that our manuscript has been accepted for publication in BMC Musculoskeletal Disorders. The manuscript has been amended to include the use of the term of face validity in the discussion as suggested by Reviewer 2.

Thank you for all your time and attention to our submission. I look forward to hearing from you with respect to the progress of this publication.

Kind regards

Dr Jeremy Horwood